Author's response to reviews

Title: Colon Capsule Endoscopy compared to Conventional Colonoscopy under routine screening conditions

Authors:

Julia B Pilz (julia.pilz@ksli.ch)
Susanne Portmann (susanne.portmann@stud.unibas.ch)
Shajan Peter (shajan@uab.edu)
Christoph Beglinger (Beglinger@tmr.ch)
Lukas Degen (ldegen@uhbs.ch)

Version: 3  Date: 9 February 2010

Author's response to reviews: see over
Basel, February 9th, 2010

Dear Dr. Edmunds

enclosed you will find a revised version of our manuscript for re-evaluation. The comments of the reviewers were helpful and did improve the quality of our manuscript. In fact, the previous version of the Results and Discussion section was in part confusing. We have therefore largely rewritten this part and we are convinced that the new version presents a clearer picture. Attached is also a detailed response to the queries raised by the reviewers. Thank you for considering the manuscript for publication in BMC.

All the best,

Julia Pilz